Massachusetts Institute of Technology Department of Electrical Engineering and Computer Science

6.033, Spring 1999 Handout 3{ Recitation Preference Form February 3

Please fill out this form in print. Return it at the end of today’s lecture. Check the 6.033 web page after 10PM tonight for your new recitation. Remember that teammates for design project 2 must have the same recitation instructor. If you do not turn in this sheet, you will not be in the class. If we are oversubscribed, students will receive priority if they preregistered, take 6.033 for credit, and have taken 6.004 (or equivalent).

Last name: First name:

Athena username: @mit.edu

Phone:
Course (circle one): 6-1 6-2 6-3 6-P (M.Eng) 18-C Other (specify)
Year (circle one): 1 2 3 4 5 G Other (specify)
Did you preregister (circle one)? YES NO
Have you taken 6.004 (circle one)? YES NO Equivalent at another university
Other computer science subjects taken:

Clearly write a large X in the boxes below for which you absolutely cannot attend. Write the numeral “1” in your first choice. Write the numeral “2” in your second choice.

(#1) TR 10AM 34-303 (#11) TR 10AM 36-155 Sussman/Rahul Rudolph/Forbess (#2) TR 11AM 34-303 (#6) TR 11AM 34-302 Sussman/Fu Rudolph/Rahul (#5) TR 12PM 34-303 Hari/Fu (#8) TR 1PM 34-302 (#7) TR 1PM 34-303 Hari/Shlyakhter Saltzer/Hu (#10) TR 1PM 36-153 (#3) TR 1PM 36-839 Teller/Boyapati Seshan/Forbess (#4) TR 2PM 36-839 (#9) TR 2PM 34-303 Seshan/Shlyakhter Saltzer/Boyapati (#12) TR 2PM 36-153 Teller/Hu

The optional 6.033 lab has limited enrollment. To apply, answer the following:

Are you already skilled at using C (circle one)? YES NO
Can you attend all lab recitations at 2PM on Fridays (circle one)? YES NO